Author's response to reviews

Title: RFE based chondroplasty in wrist arthroscopy indicates high risk for chondrocytes especially for the bipolar application

Authors:

Michaela K.I. Huber (michaela1.huber@ukr.de)
Christoph Eder (christoph.eder@ukr.de)
Markus Loibl (markus.loibl@ukr.de)
Johannes Zellner (johannes.zellner@ukr.de)
Arne Berner (arne.berner@ukr.de)
Richard Kujat (richard.kujat@ukr.de)
Michael Nerlich (michael.nerlich@ukr.de)
Sebastian Gehmert (sebastian.gehmert@ukr.de)

Version: 2
Date: 28 August 2014

Author's response to reviews:

1. The title page is revised
2. We have read the editorial policies
3. Competing interest are added
4. Authors contribution is added
5. Acknowledgements is added
6. Section titles are corrected, list of abbreviations is added